Author’s response to reviews

Title: Biochemical and histological alterations induced by the smoke of allethrin based mosquito coil on mice model

Authors:

Abdulla Al Mamun (mamungeb26@gmail.com)
Ataur Rahman (ataurm1@gmail.com)
Habibur Rahman (habib.geb.ru@gmail.com)
Faisal Hoque (kmfhoque03_gen@yahoo.com)
Zennat Ferdousi (ferdousi04@yahoo.com)
Abu Reza (rezaru@gmail.com)
Nurul Matin (nmatin2@yahoo.com)

Version: 2 Date: 20 May 2017

Author’s response to reviews:

1. All of the language mistakes were corrected as possible as by indigenous experts.

2. The change in authors and related form (author change form) will be sent to editorial office.